Title: Meet the Researchers: an alternative method of engaging patients with research in mesothelioma.

Authors:
Kate Mary Hill (k.m.hill@leeds.ac.uk)
Mags Portman (mags.portman@gmail.com)
Zsuzsanna Tabi (TabiZ@cardiff.ac.uk)

Version: 2 Date: 30 Aug 2018

Author’s response to reviews:

Reviewer #1
Thank you for the opportunity to review this revised manuscript. I believe you have done an excellent job taking account of all the comments and the paper is now exceptionally clear and useful. One very minor comment, I think a word is missing on page 20, ln 58, perhaps 'how it is'.

Response:
Thank you for spotting this typographic error: it has been corrected as you suggested and is highlighted yellow in the text.

I noticed a similar error in the acknowledgement section and have also corrected that and marked it in the same way.